

This is a peer-reviewed, final published version of the following document and is licensed under Creative Commons: Attribution 4.0 license:

Abdallah-Ou-Moussa, Soukaina, Wynn, Martin G ORCID: 0000-0001-7619-6079, Kharbouch, Omar and Rouaine, Zakaria (2024) Digitalization and Corporate Social Responsibility: A Case Study of the Moroccan Auto Insurance Sector. Administrative Sciences, 14 (282). pp. 1-27. doi:10.3390/admsci14110282

Official URL: https://doi.org/10.3390/admsci14110282 DOI: http://dx.doi.org/10.3390/admsci14110282 EPrint URI: https://eprints.glos.ac.uk/id/eprint/14507

Disclaimer

The University of Gloucestershire has obtained warranties from all depositors as to their title in the material deposited and as to their right to deposit such material.

The University of Gloucestershire makes no representation or warranties of commercial utility, title, or fitness for a particular purpose or any other warranty, express or implied in respect of any material deposited.

The University of Gloucestershire makes no representation that the use of the materials will not infringe any patent, copyright, trademark or other property or proprietary rights.

The University of Gloucestershire accepts no liability for any infringement of intellectual property rights in any material deposited but will remove such material from public view pending investigation in the event of an allegation of any such infringement.

PLEASE SCROLL DOWN FOR TEXT.





Article Digitalization and Corporate Social Responsibility: A Case Study of the Moroccan Auto Insurance Sector

Soukaina Abdallah-Ou-Moussa¹, Martin Wynn^{2,*}, Omar Kharbouch¹ and Zakaria Rouaine¹

- ¹ Faculty of Economics and Management, Ibn Tofail University, Kenitra B.P 242, Morocco; soukaina.abdallahoumoussa@uit.ac.ma (S.A.-O.-M.); omar.kharbouch1@uit.ac.ma (O.K.); rouainezakaria@gmail.com (Z.R.)
- ² School of Business, Computing and Social Sciences, University of Gloucestershire, Cheltenham GL50 2RH, UK
- * Correspondence: mwynn@glos.ac.uk

Abstract: The aim of this article is to explore the impact of digitalization on corporate social responsibility (CSR) in the automobile insurance sector in Morocco. This article first explores the theoretical and conceptual foundations of digital transformation and CSR. A mixed methods approach is then used, combining qualitative interviews with a wider quantitative survey, to investigate how digital innovations influence CSR practices. Interview analysis provides the basis for the development of a conceptual framework and eight hypotheses, which are then tested using quantitative techniques to analyze survey data. The results reveal several links between the benefits of digitalization and CSR. Claims management platforms, digital roadside assistance tools, and digital vehicle assessment and inspection all positively impact policyholders' well-being in terms of compensation and asset preservation, thereby enhancing the CSR profile of automobile insurers. Similarly, augmented reality (AR) and virtual reality (VR) training and simulation, as well as repair assistance, have positive impacts on policyholders' well-being and advance the CSR positioning of automobile insurers. This article has limitations as it is based on a narrow industrial sector in a single country, but it nonetheless highlights certain relevant interrelationships between digitalization and CSR, contributing to the development of theory and practice in these research areas.

Keywords: digitalization; corporate social responsibility; CSR; automobile insurance sector; claims management; digital technologies

1. Introduction

There is no universally agreed upon definition of digitalization, but here it is taken to mean the integration of digital technologies into organizational processes to transform data, automate operations, and improve efficiency. This aligns with the definition put forward by SAP (2024), which states that digitalization occurs "when data from throughout the organization and its assets is processed through advanced digital technologies, which leads to fundamental changes in business processes" (para. 9). This has brought welldocumented benefits to various industries, including logistics (Richert and Dudek 2023), finance (Pangalos 2023), and many aspects of manufacturing (Wynn 2021). In the financial sector, insurance companies have not been excluded from the impacts of digitalization, as practices and operational processes have been re-engineered. The industry is built around collecting, processing, and analyzing data on economic activity, relying on the prediction of future events to ensure appropriate decision-making (Ciarli et al. 2021). This approach is based on an ethical code that requires insurance companies to act responsibly toward their clients, respecting their privacy and protecting their personal data (Quach et al. 2022).

Insurance companies are responsible for covering potential risks affecting individuals' lives and properties in exchange for a financial contribution—commonly referred to as an insurance premium. The integration of digital technologies into insurance companies'



Citation: Abdallah-Ou-Moussa, Soukaina, Martin Wynn, Omar Kharbouch, and Zakaria Rouaine. 2024. Digitalization and Corporate Social Responsibility: A Case Study of the Moroccan Auto Insurance Sector. *Administrative Sciences* 14: 282. https://doi.org/10.3390/ admsci14110282

Received: 15 September 2024 Revised: 25 October 2024 Accepted: 31 October 2024 Published: 2 November 2024



Copyright: © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). operating systems and processes offers greater transparency, which in turn is likely to make corporate practices more responsible toward social and environmental stakeholders, thereby strengthening corporate social responsibility (CSR) strategies and activities (Alieva and Powell 2023). Digitalization can thus be seen as the cornerstone of strengthening CSR within companies in terms of the involvement of stakeholders and a responsibility toward them (Zhang and Liu 2022). However, other authors have pointed out that poor management of the digitalization process can exacerbate social and environmental issues and hinder economic growth within companies (Chen et al. 2023).

Xin et al. (2022) investigated how CSR can underpin social and environmental goals in such a way as to support the financial performance of insurance companies (Xin et al. 2022). Although companies' perceptions of CSR dimensions may vary across different sectors (Dudek et al. 2023), CSR has clearly become an indispensable concept in insurance company management (Ruiter 2022), generating positive gains in terms of company performance (Kavitha and Anuradha 2016). One dimension of CSR in the insurance industry involves providing financial assistance based on the principle of risk sharing in the event of unforeseen events. It also involves offering expertise to policyholders to help mitigate the consequences of environmental disasters, promote human rights, and ensure societal safety (Ruiter 2022). Strengthening CSR among insurers can contribute to ensuring an upward trend in sustainable growth and, consequently, continued economic development in the insurance sector, whilst at the same time exhibiting socially responsible strategies (Khovrak 2020).

However, there are relatively few studies that examine the links between digitalization and CSR. In the context of the textile and clothing industry, Wiegand and Wynn (2023) demonstrated how digitalization promotes transparency in supply chains, while Eling and Lehmann (2018) explored the impact of digital technologies on sustainability practices in the financial sector. The literature concerning the interaction between digitalization and CSR in the insurance sector is similarly scant. This study aims to address this gap in the current literature and, more specifically, investigates the extent to which the digitalization of insurance companies' operational processes, particularly in the automobile insurance sector, enhances the implementation of CSR, especially concerning the social dimension focused on policyholders (clients). This study assesses the impact of a set of factors related to digitalization on the adoption and perception of CSR among insurers and policyholders and contributes to the current literature by offering an innovative conceptual framework for understanding the impact of digitalization on CSR, particularly in the insurance sector. From a practical perspective, the results offer specific recommendations for insurers regarding the integration of digital technologies into their CSR strategies, thereby facilitating better claims management and a more responsive assistance process.

This article consists of six sections. Following this introduction, Section 2 briefly reviews and reports the relevant literature, a conceptual framework of digital determinants is established, and eight hypotheses are set out. Section 3 then details the research methodology and techniques used to analyze qualitative (interviews) and quantitative (survey) data. Section 4 reports the main results of hypothesis testing, and Section 5 discusses a series of emerging issues relevant to the overall research objective. Finally, Section 6 summarizes the main aspects of the study, discusses limitations, and highlights possible areas for future research in this field.

2. Literature Review and Hypotheses Development

This section explores the relevant literature pertaining to our research aims and objectives, focusing on the interconnections between digitalization and CSR in the car insurance industry. Firstly, the two primary domains investigated in this research—digitalization and corporate social responsibility—are reviewed, and the definitions of these concepts used in the existing literature are briefly discussed. Section 2.2 then examines the impact of these concepts on the insurance sector. This is followed by a discussion of some of the theories and frameworks used in this research area, and the final Section 2.4 identifies the research gap, sets out a conceptual framework based on digital determinants, and develops eight related hypotheses.

2.1. Digital Transformation and CSR

There are different perspectives and definitions of the terms "digitalization" and "digital transformation." As van Tonder and Petzer (2018) observe, "there is no solid and universally accepted conceptual framework that can help companies, practitioners, and academics understand the constructs of digitalisation, digital transformation, and business model innovation" (p. 112). Some authors consider digitalization as the deployment of digital technologies to support the enhancement of existing processes, while digital transformation is sometimes seen as constituting a more significant transition towards a new business model or, at the very least, a new way of working in a substantial part of a business.

Most authors recognize the main dimensions of CSR as being environmental, ethical, philanthropic, and economic, even though, in the digital age, new sets of responsibilities are becoming increasingly important. In this context, investment in "intangible capital"—the development of employees' skills and capabilities—is essential for successfully adapting to continuous technological change (Chien et al. 2021). Digitalization involves moving from an offline logic to online exchange, with the conversion of analog data into a digital format (Van Veldhoven and Vanthienen 2023), and this requires the development and adoption of a new digital culture, something which is increasingly seen as an element of CSR (Ahmad et al. 2024). Indeed, some authors also view corporate digital responsibility (CDR) as a new component of CSR (Wynn and Jones 2023).

Many organizations now place significant importance on social responsibility towards employees and stakeholders as part of their CSR strategy (Ahmad et al. 2024; Hernández et al. 2020). Social innovation, conflict mitigation, improved working conditions, and increased organizational competitiveness are the results of well-thought-out CSR policies (Murray et al. 2021). CSR not only advances the social prerogatives of employees and company partners, but also improves the environmental and economic prerogatives. This responsibility covers employees, customers, suppliers, shareholders, investors, and all partners operating within the company's business network (Carroll and Shabana 2010). The need for digital responsibility and the alignment of business practices with sustainability principles illustrates the close link between CSR and digitalization (Janiszewska-Kiewra et al. 2024; Yokoi et al. 2023). The digitalization of operational activities can underpin CSR strategies by improving communication and information exchange between employees at different hierarchical levels and with partners involved in the company's supply chain (Martínez-Caro et al. 2020). The effective integration of these concepts into an overall business strategy can improve the employee experience and enhance the environmental and economic performance (Esposito and Ricci 2021).

Closely related to CSR reporting and strategy development is 'environmental, social and governance (ESG)', which is a framework used to monitor business performance in these three areas of activity and is sometimes used to assess a company's sustainability. Zhang and Huang (2024) point out that, although some scholars have examined the relationship between digital transformation and corporate ESG performance, there is a lack of detailed research into the mechanisms underpinning such change. The authors conclude that "understanding how various circumstances influence the effectiveness of digital transformation in achieving ESG goals can provide companies with more nuanced guidance as they plan and implement their digital development strategies" (p. 2).

In this context, it must be recognized that whilst digitalization has some positive environmental impacts, there are also negatives. On the one hand, the use of dematerialized platforms and the rise of artificial intelligence can optimize the use of natural and energy resources (Wagner 2015). The United Nations Environment Programme (2021) concluded that "a digital ecosystem of data platforms will be crucial to helping the world understand and combat a host of environmental hazards, from air pollution to methane emissions"

(para. 5). On a more local scale, digitalization can strengthen the bonds between employees and their company, help in the development of new skills, and reduce physical work and travel, thereby reducing the carbon footprint of companies. On the other hand, as observed by DataCamp (2023), "an increasing number of studies are alerting us to the significant climate and environmental impact of our digital activities" (para. 2). The data centers upon which digitalization depend, which are major energy users and thus contribute greenhouse gas emissions and climate change, are examples of this.

2.2. Digital Transformation and CSR in the Insurance Sector

The deployment of digital technologies can positively impact the value chain of insurance providers through the improved assessment of unforeseen future events, which in turn improves the insurability of probable risks affecting individuals' lives and properties (Eling and Lehmann 2018). Digitalization can transform operational methods and risk assessment processes to reduce the likelihood of insolvency among insurers. In addition, the digitalization of operational processes in the insurance sector has produced innovative working practices that enhance sustainability and compliance with environmental standards (Ehrentraud et al. 2020).

Digitalization impacts various insurance functions, enabling improved pricing processes, policy underwriting, and claims management when insured risks materialize (Mizgier et al. 2018). The digitalization of operational processes can reduce service and processing costs, adapt pricing to potential risks when underwriting insurance policies, mitigate human errors, and expand the insured base, allowing reinvestment in other innovative products (Porter and Heppelmann 2017). Claims management processes have been enhanced, and fraud detection has been strengthened (Garde and Prasad 2018).

Insurance companies are increasingly integrating digital tools to optimize their operations, enhance efficiency, and meet consumer expectations. These technologies include AI, blockchain, the Internet of Things (IoT), and Big Data, as well as claims management platforms, online customer portals, and digital roadside assistance. They not only reshape insurers' internal processes but also redefine interactions between businesses and their customers, directly impacting the risk management process. AI, for example, can accelerate processes and procedures, and can also improve the accuracy of forecasts and claims assessments, a key factor in remaining competitive in an increasingly digitalized market (Erem Ceylan 2022). Furthermore, integrating blockchain into insurance processes helps to reduce fraud risk while increasing transaction transparency, thereby strengthening customer trust (Popovic et al. 2020). IoT devices can enhance real-time data capture capabilities to enable the creation of more dynamic insurance products, tailored to individual customer behaviors. This technology is particularly useful in digital vehicle assessment, allowing for faster and more accurate inspections after a claim. Additionally, predictive analytics based on Big Data identifies previously invisible trends, paving the way for increased service personalization and better risk management.

The use of advanced security technologies and data analysis for personalization further enhances insurers' ability to offer customized products that meet specific customer needs. Insurers adopting these digital platforms are thus better positioned to provide personalized services and improve customer satisfaction, a crucial issue in a competitive context (Eckert et al. 2022). Personalization is also enhanced via the use of augmented reality (AR) and virtual reality (VR) to simulate and evaluate risks interactively, as well as by integrating e-commerce to simplify the purchase and management of insurance policies. In this context, the impact of these innovations on CSR becomes central. Digital technologies not only modernize the insurance sector, but they also transform CSR approaches, enabling insurance companies to better meet societal and environmental expectations. By integrating these technologies, insurers can enhance transparency, inclusion, sustainability, and social engagement, contributing to developing more robust and relevant CSR practices suitable for contemporary challenges.

2.3. Relevant Theories and Frameworks

There are a number of theories and models with relevance to this study. The dynamic capabilities theory asserts that public or private organizations must regularly channel their skills and allocate human, technical, and financial resources to adapt to continuous technological changes in order to meet society's growing expectations (Teece 2018). Simultaneously, the sustainable innovation theory posits that these new technologies lead to ongoing changes in sustainable operating modes and stimulate responsibility towards employees and partners in social, environmental, and economic areas (Schaltegger et al. 2016). These two theories are complementary, highlighting the interconnection between allocated resources, technological advancements, and operational practices. This suggests that any organization motivated by the need to adapt to an ever-changing environment must continuously invest in new technological processes to keep pace with evolving operating modes.

Beyond the traditional economic approach based on the usual production factors capital and labor—the resource-based theory identifies intangible elements in order to create value from available resources. The theory asserts that intangible resources far exceed tangible resources in importance in the value creation process within companies (Salvi et al. 2021). This reflection has been endorsed by a large part of the scientific community (for example, Haji and Mohd Ghazali 2018; Ahmad et al. 2023). The imperatives to enhance intangible resources and companies' awareness of the importance of digital culture explain the progress of CSR in terms of its various social, environmental, and economic dimensions (Ahmad et al. 2023).

In this context, Heeks's model (Heeks 2002) for information systems implementation, which examined the gap between design and actuality (the "design–actuality gap"), is of relevance. Heeks identified four dimensions of change when implementing new technologies: technology, processes, people, and structures. From a purely technological perspective, digital transformation involves the adoption of new digital technologies such as mobile technologies, advanced analytics, and augmented reality. In terms of process (or organizational) change, business processes are reshaped to create new, flexible, and consistent management practices (Hanelt et al. 2021). At the same time, the 'people' (or social) dimension of digital transition may involve re-skilling and changes in roles and responsibilities (Reis et al. 2018). Finally, the integration of digital technologies within an organization aims to establish an innovative digital management model that drives growth and value creation, which may involve structural change within the organization (Verhoef et al. 2021).

Some of the recent debate around digitalization and digital transformation builds upon the change dimensions identified in Heeks' model. As noted above, there are various perspectives in the current research literature as to what exactly constitutes digital transformation. Pratt (2023), for example, sees a digital transformation strategy as amounting to the deployment of digital technologies to continuously create new products, services, processes and engagement channels, whilst Van Alstyne and Parker (2021) highlight the role of factors and entities external to the organization. Ismail Abdelaal et al. (2018) see digital transformation as resulting in the development of a new business model with new digitally enabled products and services, while simultaneously impacting people (including skills, talent, and culture), processes, and networks. Digital transformation, then, can be differentiated from digitalization by the deployment of digital technologies in the organization's products and/or services. This is in addition to the more general role of improving cross-company business processes that epitomizes "normal" digitalization. Digital transformation, by directly changing a company's product offering and/or service provision, will lead to a more radical change in business processes, people's competencies and skills, and value networks.

In summary, the dynamic capabilities theory is particularly useful for understanding how insurers continually adapt to rapidly evolving digital technologies in order to remain competitive. This theory highlights the need for companies to acquire, combine, and reorganize their resources in the face of technological change. Also of relevance is the theory of sustainable innovation, which highlights the importance of integrating technological solutions that promote sustainability, which is crucial for CSR initiatives in the insurance sector. Finally, Heeks' model offers an approach with which to evaluate the effectiveness of digital infrastructure in promoting sustainable development. These theories will help structure the conceptual framework and contribute to the development of hypotheses.

2.4. Conceptual Framework and Hypotheses Development

Digital transformation in the insurance sector, particularly in Morocco, raises new questions about integrating CSR activities within a fast-changing technology environment. While digital technologies offer significant opportunities to improve the efficiency and transparency of insurance processes, their impact on achieving CSR objectives remains largely unexplored. Current research reveals significant gaps in evaluating how these innovations can align with CSR strategies, particularly concerning the specific needs of the automobile insurance sector. To address this gap in the literature, this research uses a conceptual framework based on digital determinants (related to specific digital technologies) of successful CSR. This framework was developed from an analysis of the extant literature, which identified the digital technologies likely to play a key role in promoting effective and adapted CSR practices. These determinants, selected for their relevance and the potential they demonstrated in terms of enhancing sustainability, fairness, and satisfaction among stakeholders in the insurance sector, are as follows: claims management platforms and online customer portals; digital roadside assistance; digital vehicle assessment and inspection; advanced security technologies; data analysis for personalization; and enhanced customer experience through augmented reality (AR) and virtual reality (VR). These are explored in more detail below.

Claims management platforms and online customer portals: A claims management platform is a digital system used by insurance companies to centralize, automate and track the claims process. These platforms help reduce processing times, minimize human errors, and improve the transparency of operations. Several studies highlight the advantages of using digital platforms for claims management. Yun and Barde (2024) conclude that the use of digital dashboards for automobile warranty management improves payment accuracy, whilst Muiru (2024) found that the automation of claims management processes improves the operational efficiency of companies and increases policyholder satisfaction.

The innovative use of digital platforms in the insurance sector thus offers various advantages. They support routine and administrative tasks, contribute to complex decisionmaking tasks, and manage operational processes. These digital platforms aim to relieve insurance organizations of operational tasks and help them to focus on strategic decisions (Nicoletti 2020). Online mobile channels present rigorous alternatives to traditional management tools, enabling mass data collection, real-time analysis, behavior-based underwriting and pricing, and agile compensation in the event of insured risks, resulting in remarkable operational efficiency (Albrecher et al. 2019). The use of online management platforms combined with the application of artificial intelligence to assess insurability allows insurance companies to accurately predict loss probabilities, reduce information asymmetry, and rethink insurance coverage to offer policies that align with risks and meet policyholders' expectations (Eling et al. 2022). In contrast to archaic methods of claims management in the insurance industry, the introduction of technologies such as blockchain strengthens the transparency, standardization, and reliability of data. The implementation of this digital process through online interfaces in claims management and compensation involves reducing operational costs and improving operational processes in terms of standardization and data reliability, demonstrating an immutable character (Gillis 2023). However, digital transition and dematerialized platforms are likely to affect the entire value chain of insurers and increase policyholder responsiveness (Eling and Lehmann 2018).

Digital roadside assistance: Digital roadside assistance can be defined as the emergency services provided to drivers through mobile platforms and applications. This involves

the use of technologies such as the Internet of Things (IoT) to improve the efficiency and responsiveness of interventions in the event of a breakdown or accident. Mattioli et al. (2024) show that the adoption of IoT-optimized roadside assistance systems enables the real-time monitoring of vehicle health and the prediction of breakdowns, contributing to faster and more efficient interventions. These systems reduce operational costs by optimizing support team journeys, and Donald and Brail (2024) found that digital platforms can effectively coordinate road services, including waiting times, and improve the quality of services provided to policyholders. Digital roadside assistance reduces wait times and improves customer satisfaction, contributing to the transparency and speed of interventions (Ren and Chen 2024). The digitalization of claims processes and the provision of digital roadside assistance to policyholders improves service delivery and enhances operational efficiency, allowing insurers to focus more on the customer experience (Kemboi 2022).

Digital vehicle assessment and inspection: This concerns the use of digital technologies, such as AI and sensors, to assess and inspect vehicle damage, often through automated systems or apps. Bianchi et al. (2024) found that digital inspection technologies improve the accuracy of assessments by minimizing human errors and speeding up decision-making processes. AI enables deeper damage analysis and more objective decision-making, thereby improving claims transparency and policyholder satisfaction. The adoption of these technologies has a direct effect in terms of reducing compensation times, and also improves customer service as a result of the speed and accuracy of assessments (Wang et al. 2024).

Indeed, the digitalization of insurance company functions and the use of artificial intelligence in their operational modes now cover the entire insurance lifecycle, particularly the compensation process, encompassing evidence collection, claims file analysis, damage assessment, compensation, and fraud detection (ChAD 2021). These technological devices fully automate the compensation process, from evidence collection to decision-making regarding claim resolution. Automated compensation relies on different data from similar previous cases to determine how to resolve a new claim and calculate the amount of compensation to be paid. This method ensures fast claim processing and a low risk of errors (IndustryWired 2021).

Advanced security technologies: These include monitoring and anomaly detection systems, such as anti-theft sensors and driver assistance devices, which improve the safety of vehicles and policyholders. Ren and Chen (2024) demonstrate that the integration of these technologies into vehicles contributes to better risk management, thus strengthening road safety and policyholder satisfaction. These innovations also reduce claims incidents, contributing to lower costs for insurance companies. The use of advanced security technologies helps to reduce claims while improving policyholder satisfaction. More generally, the integration of artificial intelligence into insurance operating processes can not only predict risks but can also mitigate them by encouraging policyholders to adopt safe behaviors (Kelley et al. 2018; ChAD 2021).

Data analysis for personalization: Data analysis allows insurers to personalize their offers based on the specific behaviors and needs of policyholders, thereby improving satisfaction and loyalty. Stanly and Aruna (2024) reported that data analysis in the insurance sector makes it possible to offer services that are more adapted to policyholders, improving their satisfaction while optimizing company offers. Data analysis improves the personalization of offers and strengthens the relationship between insurers and policyholders, which contributes to better customer satisfaction (Kelley et al. 2018; Ross 2020). These technological advances not only allow for the analysis of collected mass data but also predict and identify new needs, establish new risk profiles, and offer competitive premiums tailored to policyholders' specificities (Lustman 2021). The digitalization of operational processes and the use of artificial intelligence tools contribute to calculating the probability of risk occurrence based on policyholders' historical data to help make decisions on subscriptions and the renewal or termination of insurance contracts. The personalization of insurance services, particularly the modular services desired by policyholders, is a key competitive

action that requires the availability of data and a reduction in information asymmetry to achieve a favorable market position (Trescases 2019).

Augmented reality (AR) and virtual reality (VR) for enhanced customer engagement: AR and VR are among the interactive technologies that bridge the gap between digital and physical reality (Oyewole et al. 2024). These two emerging technologies offer strong customer engagement in both virtual and physical domains (Goyal et al. 2023) and provide a level of interaction that was previously inaccessible through traditional digital channels. The immersive nature of these devices creates a simulated environment where financial concepts and products can be visualized and understood more intuitively, leading to betterinformed decisions and higher customer satisfaction (Soni et al. 2022). As regards auto insurance, the use of AR enables customers to better understand the complex terms of insurance policies and visualize the consequences of different accident or claim scenarios. Additionally, VR can simulate virtual road environments where drivers can experience the benefits of various coverage levels in real-life conditions (Wieland et al. 2024). This innovative approach goes beyond merely presenting products; it fosters more engaging interaction and allows users to compare insurance options in real-time based on their specific profile and needs (Nicoletti 2021). In this way, companies in the insurance sector can enhance their business efficiency while improving customer satisfaction.

In summary, claims management platforms enable more transparent and efficient claims handling, thereby reducing fraud risks and improving customer satisfaction; digital roadside assistance, which provides fast and personalized help to policyholders, enhancing customer safety and trust; and digital vehicle assessment and inspection tools, which facilitate accurate and quick damage assessment, contributing to fair and data-driven decisions. Additionally, online customer portals offer policyholders centralized and simplified access to their information, improving transparency and communication between the insurer and the customer. Advanced security technologies, in turn, enhance accident and claim prevention, thereby reducing risks and better protecting policyholders. Data analysis enables personalized insurance offers, meeting specific customer needs while promoting fairness. Furthermore, augmented reality and virtual reality can be used to train policyholders and employees, improving their understanding of insurance products and associated risks. Finally, integrated e-commerce simplifies access to insurance products and associated services, streamlining the purchasing process and enhancing customer accessibility.

Integrating these technologies within the CSR framework allows insurance companies to not only improve their operational efficiency but to also proactively respond to societal and environmental expectations. This conceptual framework serves as the basis for developing the following hypotheses to test the impact of these technologies on CSR in the automobile insurance sector (Figure 1).

H1. Claims management platforms (process automation and transparent communication) positively impact policyholders' well-being in terms of compensation and asset preservation by pooling risks, thereby contributing to the success of CSR use among automobile insurers.

H2. Digital roadside assistance (assistance applications and accident help) positively impacts policyholders' well-being in terms of compensation and asset preservation by pooling risks, thereby contributing to the success of CSR use among automobile insurers.

H3. Digital vehicle assessment and inspection (virtual inspections and image and video analysis) positively impacts policyholders' well-being in terms of compensation and asset preservation by pooling risks, thereby contributing to the success of CSR use among automobile insurers.

H4. Online customer portals (self-service and historical tracking) positively impact policyholders' well-being in terms of compensation and asset preservation by pooling risks, thereby contributing to the success of CSR use among automobile insurers.

H5. Advanced security technologies (anti-theft systems and driver assistance through sensors and cameras) positively impact policyholders' well-being in terms of compensation and asset preservation by pooling risks, thereby contributing to the success of CSR use among automobile insurers.

H6. Data analysis for personalization (customized offers and rewards) positively impacts policyholders' well-being in terms of compensation and asset preservation by pooling risks, thereby contributing to the success of CSR use among automobile insurers.

H7. Augmented reality (AR) and virtual reality (VR) (training and simulation, repair assistance) positively impact policyholders' well-being in terms of compensation and asset preservation by pooling risks, thereby contributing to the success of CSR use among automobile insurers.

H8. Integrated e-commerce (the purchase of parts and services, price comparison, etc.) positively impacts policyholders' well-being in terms of compensation and asset preservation by pooling risks, thereby contributing to the success of CSR use among automobile insurers.



Figure 1. Digital technologies, digital determinants, and related hypotheses (conceptual framework).

3. Research Method

This study focuses on the intersection of digitalization and CSR in the car insurance industry (Figure 2), and more specifically investigates the car insurance sector in Morocco, examining how operational processes have been impacted by digitalization, and how this has affected CSR. Morocco has benefitted from government initiatives to promote digitalization, making it fertile ground for studying the impacts of digital innovation on corporate social responsibility (CSR) in the insurance sector. As the leading automotive player in Africa, Morocco has been able to leverage its strategic position in the Middle East and North Africa (MENA) region to modernize its financial services sector, integrating advanced technologies that meet growing consumer demands and sustainability standards. As a leader in the automobile industry in Africa, Morocco plays a crucial role in regional dynamics. Its commitment to modernizing financial services and integrating digital technologies allows it to serve as a model for other developing countries. The lessons learned from the Moroccan experience can be extrapolated to other emerging markets, thus strengthening the relevance of this study and its results.



Figure 2. Research focus: the interaction between digitalization, CSR, and the car insurance industry.

The development of a research methodology for such a project typically begins with a fact or theory. This is followed by data collection to clarify the problem in order to either accept or reject the theory. Finally, tests are conducted to complete the process (Park et al. 2020). In this project, the research centers on the automobile insurance sector in the Rabat-Salé-Kenitra region of Morocco. Positivism is adopted as the epistemological paradigm within a hypothetico-deductive reasoning framework to support the research process (Park et al. 2020; Hofmann 2022). A mixed method was adopted, combining qualitative interviews with quantitative data drawn from surveys. This approach provides a comprehensive and nuanced view of the effects of digitalization on claims management and roadside assistance processes. Whilst the overall methodology could be viewed as a form of pragmatism, the authors believe that the rigorous testing of hypotheses supports a positivist stance.

There were four main phases in this research study (Figure 3). Phase 1 entailed a comprehensive review of the pertinent literature. A qualitative approach was adopted to identify key themes related to digitalization that could also potentially advance sustainability in the insurance sector. Content analysis was used to extract and categorize key themes from the relevant documentary sources, focusing on recurring themes and important variables. This produced an initial list of ten key themes, which were termed "digital determinants". Then, in Phase 2 of the study, a questionnaire was developed based on the digital determinants identified in the literature review, and this was used in direct face-to-face interviews with over 100 policyholders and insurers in the automobile sector to gather their views on the impact of the digital determinants in terms of the possible enhancement of CSR amongst insurers.

Using multiple correspondence analysis (MCA) (Ge and Whitmore 2010; Adwere-Boamah and Hufstedler 2015), data were analyzed to identify those digital determinants that had a highly significant impact on the success of CSR in the automobile insurance branches. MCA is similar to principal component analysis in that it is used to analyze the pattern of relationships between several qualitative dependent variables (Abdi and Valentin 2007). This method is particularly suited to the study of correspondences between several qualitative variables, making it possible to visualize the interrelations between the factors identified from the interviews and the data collected via the survey. Statistical analyses, including logistic regression and AUC (area-under-the-curve) tests, were performed to measure the impact of digital determinants on CSR success in the auto insurance sector (see Appendix D). Reliability tests and chi-square tests were also undertaken, and the results confirmed the existence of a highly significant association between explanatory variables (such as claims management platforms and advanced security technologies) and CSR success, with *p*-values of less than 0.05. This approach not only made it possible to validate the proposed hypotheses, but also contributed to better understanding the specific



Figure 3. Research process of this study.

This led to a reassessment of the significance of the digital determinants, with four being rejected on the basis of interview feedback, leaving a remaining core of six digital determinants. Phase 2 also facilitated the development of hypotheses built around the remaining six digital determinants. Overall, eight hypotheses were generated for testing in Phase 3.

Phase 3 of the research used an online survey (Liu and Jung 2021) to further refine the digital determinants and test the hypotheses. The online survey was set up to estimate the impact of each previously identified digital determinant on the success/failure response variable of the CSR concept with regard to insurance companies. The participants were selected according to precise inclusion criteria. We selected policyholders who had taken out a car insurance contract for at least one year in the regions of Rabat, Salé, and Kénitra. Participants who did not meet these criteria were excluded to ensure the representativeness of the sample. The exclusion criteria were clearly defined to avoid any bias in the responses.

In the survey, the introductory section focused on the respondent's (policyholder's) personal information and their relationship with their insurer (the gender of the policyholder, age group, city of residence, socio-professional category, the insurance companies contracted by the policyholder, the types of insurance contracts, and the frequency of automobile insurance subscription). The second section was dedicated to responses to multiple-choice questions related to the digital determinants explaining the dependent variable. These were assessed using a Likert scale. In the final section, the survey contained questions regarding the perceived success or failure of CSR. We contacted a sample of 1100 automobile insurance policyholders, and 1000 valid survey responses were obtained. The demographic profile of the 1000 respondents is included as Appendix A. The data were analyzed to identify the digital determinants with a very significant impact on CSR success in the automobile insurance sector.

Phase 4 used the survey data to test the hypotheses, applying generalized linear models (GLMs), particularly the logit extension and the binary logistic regression models. This made it possible to model the impact of numerical determinants on the success or

the failure of CSR (Banerjee et al. 2024; Kumar and Gota 2023) (see Appendix B). The survey used a 5-point Likert scale to assess respondents' perceptions of the impact of digital technologies on CSR. The survey items (questions, scales, etc.) were carefully developed from the existing literature, including the work of Ge and Whitmore (2010) and Adwere-Boamah and Hufstedler (2015), and were adapted to align with the specificities of the insurance sector in Morocco. The complete results of the reliability analysis, such as Cronbach's alpha ($\alpha = 0.881$), show satisfactory internal consistency for the six numerical determinants studied.

The GLMs use the logit function to model the log odds of an event as a linear combination of independent variables. This method provides the binary response variable (success/failure) of the CSR concept in the automobile insurance sector (Leukel et al. 2024). The logit statistical model was used to evaluate the impact of the digital determinants, identified from the interview analysis, on the success of CSR among automobile insurers. The β parameters are the coefficients that reflect the influence of explanatory variables on the dependent variable in logistic regression. They allow for the measurement of the effect of digital determinants on the probability of CSR success in the automobile insurance sector. To estimate these parameters using the nonlinear equations of the Bernoulli distribution, the maximum likelihood estimator (MLE) was employed (see Appendix C).

In summary, the research methodology combined epistemological rigor and robust statistical tools, constituting a solid basis for analyzing the impact of digital determinants on the success of CSR in the automobile insurance sector. This integrated approach, based on the combination of qualitative and quantitative methods, not only ensured the validity of the results obtained but also their applicability to the specific context of the Moroccan market. The application of binary logistic regression allows for a more in-depth examination of the complex relationships between the variables studied, providing relevant insights into the dynamics underpinning the effectiveness of CSR in this sector.

4. Results

This section presents the main findings of the hypothesis testing, derived from the statistical analysis of the data collected through interviews and online surveys. The results are discussed in relation to the digital determinants identified in the conceptual framework (Figure 1) and their impact on the success of CSR uptake among automobile insurers in Morocco. The subjects in the sample chose to take out their automobile insurance contracts with a number of different insurance companies, most of them well-known brands (Table 1). To ensure the rigor of the results, we assessed the reliability of the quantitative data using Cronbach's alpha ($\alpha = 0.85$), with results indicating high internal consistency. The validity of the results was confirmed by convergent and discriminant validity tests. An exploratory factor analysis was conducted to identify the key dimensions of digital determinants that influence CSR. Logistic regression was then used to test the hypotheses regarding the relationships between these dimensions and the success of CSR initiatives.

	Respondents	Percentage	Cumulative Percentage
Allianz Maroc	26	2.6%	2.6%
Atlanta	45	4.5%	7.1%
AXA Assurance	114	11.4%	18.5%
MCMA	63	6.3%	24.8%
RMA	152	15.2%	40.0%
Saham Assurance	306	30.6%	70.6%
Sanad	49	4.9%	75.5%

Table 1. Insurance companies contracted by respondents.

	Respondents	Percentage	Cumulative Percentage
Wafa Assurance	179	17.9%	93.4%
Others	66	6.6%	100%
Total	1000	100%	-

Table 1. Cont.

The automobile insurance policies taken out were of three main types: civil liability (16.8%); civil liability + additional guarantees (damage and collisions) (61.2%); and fully comprehensive (22%). The contract subscription frequencies varied between annual policies (55.7%); half-yearly policies (12.8%); quarterly policies (23.6%); monthly policies (7.3%); and other policies (0.6%). More detail on the hypotheses testing method and validation is given in Appendix D.

4.1. Claims Management Platforms and Online Customer Portals (H1, H4, H8)

The analysis confirms that claims management platforms and online customer portals have a significant positive impact on the success of CSR in the automobile insurance sector. These digital platforms enhance operational efficiency by automating processes, reducing human errors, and ensuring transparency in communication. Policyholders benefit from more streamlined claims processing, quicker compensation, and greater accessibility to their insurance information, all of which contribute to their well-being and satisfaction.

- H1 is supported: the automation and transparency offered by claims management platforms positively influence policyholders' well-being by ensuring timely and fair compensation, which aligns with CSR objectives.
- H4 is supported: online customer portals improve policyholder engagement and trust by providing easy access to information and self-service options, thereby supporting CSR efforts.
- H8 is supported: the integration of e-commerce functionalities in these platforms enhances service personalization and accessibility, which further contributes to CSR success.

4.2. Digital Roadside Assistance (H2)

The findings indicate that digital roadside assistance significantly improves policyholders' perception of their insurance providers' responsiveness and reliability. The availability of instant, personalized assistance in the event of a breakdown or accident increases policyholder trust and loyalty, which is crucial for maintaining strong CSR practices.

• H2 is supported: digital roadside assistance positively impacts policyholders' wellbeing by providing timely and effective help during emergencies, thereby supporting CSR in the insurance sector.

4.3. Digital Vehicle Assessment and Inspection (H3)

This study reveals that digital vehicle assessment and inspection tools contribute to faster and more accurate damage evaluations, leading to fairer and more transparent compensation processes. This not only enhances policyholder satisfaction but also strengthens the insurer's CSR profile by ensuring that claims are handled equitably and efficiently.

 H3 is supported: The use of digital tools for vehicle assessment and inspection positively impacts policyholders' well-being by ensuring accurate and timely compensation, which aligns with CSR goals.

4.4. Advanced Security Technologies (H5)

The research confirms that advanced security technologies, such as anti-theft systems and driver assistance tools, play a critical role in preventing accidents and minimizing risks.

These technologies not only protect policyholders' assets but also promote safer driving behaviors, which is a key component of CSR.

 H5 is supported: advanced security technologies positively impact policyholders' well-being by reducing the likelihood of accidents and thefts, thereby contributing to the success of CSR initiatives.

4.5. Data Analysis for Personalization (H6)

The results highlight the importance of data analysis in personalizing insurance services. By leveraging Big Data and AI, insurers can offer customized policies that better meet individual policyholders' needs, leading to higher satisfaction and engagement. This personalization is crucial for maintaining a strong CSR stance, as it demonstrates a commitment to meeting the specific needs of customers.

 H6 is supported: data-driven personalization positively impacts policyholders' wellbeing by offering tailored insurance solutions that align with their specific needs, supporting CSR objectives.

4.6. Augmented Reality (AR) and Virtual Reality (VR) for Enhanced Customer Engagement (H7)

This study finds that AR and VR technologies significantly enhance policyholder education and engagement by providing immersive experiences that simplify complex insurance concepts. This not only improves policyholders' understanding but also builds trust in the insurer, which is essential for the successful adoption of CSR.

 H7 is supported: the use of AR and VR in training and customer engagement positively impacts policyholders' well-being by enhancing their understanding of insurance products, thereby supporting CSR efforts.

The qualitative interviews revealed several key themes, including the improvement of transparency and communication using digital tools. These qualitative results complement the quantitative data and offer additional insights into the mechanisms through which digitalization influences CSR. This is discussed in more detail below.

5. Discussion

The above results raise some issues worthy of further discussion. Firstly, the findings underline the interconnection of digitalization and CSR policies and operations. Digital technologies typically operate alongside standard information systems for processing and reporting transactions, and help insurers reduce production costs by limiting risks to policyholders and increasing their satisfaction levels (Bohnert et al. 2019; Hirsch-Kreinsen 2020). The dematerialization of insurance services (such as through the transition from physical to electronic certificates) and the disintermediation of operational processes (removing activities to improve process efficiency) have improved customer access to insurance services and also encouraged the financial inclusion of the wider population (Eling and Lehmann 2018). Digitalization is now integrated into the insurance sector as a key element of the overall strategy, enhancing risk assessment and creating new products tailored to policyholders' expectations (Stoeckli et al. 2018). AI and analytics, combined with Big Data, enable insurers to analyze and predict potential risks and offer insurance services tailored to different social strata, thereby promoting socially responsible behavior towards policyholders (Merrill et al. 2019).

Digitalization can contribute to reducing environmental footprints and ensuring sustainable development by minimizing physical transactions and promoting remote work (Eckert et al. 2021). This again highlights the interdependence between digital transition and CSR: the continued evolution and advancement of CSR in its multiple dimensions depends on innovation and the application of digital technologies. To this end, continuous investment in digital technologies is necessary to maintain a competitive edge and ensure the alignment of these technologies (Cirillo et al. 2023) with CSR objectives in order to guarantee long-term sustainability. This aligns with the dynamic capabilities and sustainable innovation models noted above in Section 2.3. Secondly, the results of this study confirm the importance of digital determinants for the success of CSR in the automobile insurance sector, supporting several of the original hypotheses, with the findings being of relevance to the theoretical underpinnings of this study. For example, the results indicate that the automation offered by claims management platforms has a significant impact on policyholder satisfaction (OR = 9.034-i.e., Odds Ratio in the logistic regression), confirming the importance of having a dynamic capability to adapt processes to changing needs. This is consistent with dynamic capabilities theory, which suggests that organizations must mobilize their resources to respond to constantly changing external conditions (Teece et al. 1997). At the same time, the results show that the integration of advanced security technologies reduces risks for policyholders while promoting more responsible behaviors (OR = 7.242). These innovations contribute to the sustainability of insurance processes, supporting the theory of sustainable innovation, which highlights the importance of alignment between technological innovation and sustainable practices (Schaltegger and Wagner 2011). In connection with the theory of service personalization, this study indicates that the personalization of insurance offers not only strengthens policyholder satisfaction, but also contributes to more responsible practices, whilst at the same time improving customerinsurer relationships. This is consistent with work on customer satisfaction and social responsibility (Bendell 2005).

In this context, insurers are increasingly focusing on enhancing the customer experience through personalized services and user-friendly digital platforms (Bilgihan et al. 2016), aligning with CSR objectives to improve customer satisfaction, trust, and transparency. This customer-centric approach not only strengthens relationships with policyholders but also contributes to the broader CSR goals of fairness and equity in service delivery. However, the challenge lies in ensuring that these digital innovations genuinely meet customers' needs and are not merely used as marketing tools (Verma and Bala 2018). It is, therefore, crucial to continually evaluate and adjust digital strategies based on customer feedback and evolving expectations in order to maximize their positive impact on CSR outcomes.

Thirdly, based on the case study evidence, and given the two points discussed above, the impact of digital technology deployment in the car insurance industry can be viewed as transformative, in that digital technologies are used not just to support and modernize existing business processes, but are also now an intrinsic part of the industry's products and customer service provision. As noted above, this distinction between the two concepts has been made by a number of authors (Ismail Abdelaal et al. 2018; Wynn and Felser 2023), with digital transformation involving a more significant change process than digitalization. In this study, the transformative elements include the use of mobile apps for customer roadside assistance, personalized access to online portals, and online policy management services. Company skills and competencies have had to adapt, with new data analytical skills and VR/AR capabilities adopted, whilst process change has included virtual vehicle inspections and automated claims management (Figure 4).

The β coefficients represent the strength of the relationship between the independent variables and the dependent variable (CSR success).

Fourthly, as regards the theoretical contribution of this research, the above discussion indicates how the findings extend the dynamic capabilities theory, with the insurance companies having to continuously adapt to technological change to improve operational efficiency and policyholder satisfaction. In addition, this research contributes to sustainable innovation theory by showing how digital tools facilitate transparency, personalization, and responsible risk management practices. This dual contribution—strengthening CSR practices while enhancing operational processes—provides fresh insights into the role of digital transformation in shaping socially responsible business models within the insurance industry. In a wider context, the results also enhance the theoretical understanding of the relationship between digitalization and CSR, particularly in the context of the automobile insurance sector. It builds upon the limited body of literature on this subject (Eling and

Lehmann 2018; Kong and Liu 2023) by investigating how specific digital innovations positively affect CSR outcomes. During logistic regression, we obtained β coefficients for each relationship tested between the independent variables and the dependent variable, and these are shown in Figure 4. This indicates that automated claims management and advanced security technologies had the most positive impact on customer perception of CSR, but that other factors were also of significance. These findings provide the basis for further research and theoretical development regarding this relationship.



Figure 4. Digital transformation and CSR in the car insurance industry.

Fifthly, the integration of digital technologies into CSR strategies presents challenges related to change management within insurance companies. While digital technologies offer substantial advantages, their implementation, as noted above, will often disrupt existing processes and face resistance from employees (Zuperkienė et al. 2023). Effective change management is thus critical in order to ensure that digital initiatives are seamlessly integrated into operations without compromising productivity or CSR goals. In addition to investing in new technologies, insurers must prioritize staff training and support help them in terms of adapting to these changes, fostering a culture of innovation (Chaudhuri et al. 2023) aligned with CSR objectives. Successful digital transformation relies not only on technology but also on the organization's ability to embrace change in a way that strengthens its commitment to social responsibility (Varshney 2020). This aligns with Heeks' (2002) model, emphasizing the connectivity between the technology, people, and process dimensions of digital transformation. Sixthly, this study reveals the critical importance of addressing privacy and data security concerns within the context of digital transformation. With increasing reliance on artificial intelligence, Big Data, and digital platforms, protecting sensitive customer data becomes paramount (Asif et al. 2024). Insurers must implement robust cybersecurity measures and transparent data management practices to safeguard this information and maintain trust (Stewart 2023), which is a fundamental pillar of CSR. Compliance with data protection regulations, such as the GDPR, is essential, as noncompliance could result in legal, financial, and reputational risks (Voigt and Von Dem

Bussche 2017). Ensuring high standards of data security not only mitigates risks but also strengthens the insurer's ethical commitment to responsible business practices.

Finally, the environmental impact of digitalization must be considered as insurers adopt increasingly digital operational models. While digital technologies can reduce the use of physical resources—such as paper and in-person services—they also lead to increased energy consumption by data centers and electronic devices (Chen et al. 2020). It is essential to find a balance between the environmental benefits of digitalization and the associated energy needs for insurers aiming to align with global sustainability efforts. Investment in green technologies and energy-efficient practices can help to mitigate the environmental footprint of digital transformation (Feroz et al. 2021), thereby reinforcing insurers' commitment to sustainable development and combating climate change.

6. Conclusions

This study explores the complex relationship between digitalization and CSR in the Moroccan automobile insurance sector. The findings confirm that digital technologies, such as claims management platforms, digital roadside assistance, and advanced security systems, significantly contribute to the success of CSR initiatives by enhancing policyholder well-being, engendering transparent and fair processes, and promoting safer behaviors. The study's findings are consistent with those of Eling and Lehmann (2018), who also observed that the adoption of digital technologies leads to improved sustainability practices. However, this study explores the specific impacts of this on claims management and roadside assistance, two key areas for insurers. This study thus has significant theoretical implications, broadening the field of research on the interaction between digitalization and CSR, and demonstrating the relevance and application of dynamic capabilities and sustainable innovation theories. The findings also contribute to the debate on digital transformation and how it differs from digitalization; namely, products and services are significantly modified or reinvented. From a practical point of view, the article provides guidance for insurers seeking to improve their CSR practices through the adoption of digital platforms, with benefits in terms of transparency and customer satisfaction.

While the study provides valuable insights, it has its limitations. The research is confined to a specific sector within a single country, which limits the generalizability of the findings. Additionally, the rapidly evolving nature of digital technologies means that the results may become outdated as new innovations emerge and are embedded in the insurance industry processes. Nevertheless, the authors believe that the findings provide a platform for further research into the relationship between digital transformation and CSR, and that the study also makes a small contribution to the developing literature and theory on digital transformation.

Future research could explore the impact of digitalization on CSR in other sectors and countries. The approach of testing the relationship between selected digital determinants and CSR could be applied in other industries. Additionally, longitudinal studies could provide a deeper understanding of how the relationship between digitalization and CSR evolves over time. Further research could also investigate the role of emerging technologies, such as blockchain and AI, in shaping the future of CSR in the insurance industry.

In conclusion, this study highlights the critical role of digitalization in enhancing CSR within the Moroccan automobile insurance sector. By adopting innovative digital solutions, insurers not only improve operational efficiency but also strengthen their commitment to social responsibility, ultimately contributing to a more sustainable and customer-centric insurance industry. Such a transformation, however, requires a significant change in business model, involving new products and services that require major reskilling and competency development.

Author Contributions: Conceptualization, S.A.-O.-M., M.W., O.K. and Z.R.; methodology, S.A.-O.-M., O.K. and Z.R.; software, S.A.-O.-M. and Z.R.; validation, S.A.-O.-M., M.W., O.K. and Z.R.; formal analysis, S.A.-O.-M., M.W., O.K. and Z.R.; investigation, S.A.-O.-M.; resources, S.A.-O.-M., O.K. and Z.R.; data curation, S.A.-O.-M., M.W., O.K. and Z.R.; writing—original draft preparation, S.A.-O.-M.,

18 of 27

M.W., O.K. and Z.R.; writing—review and editing, S.A.-O.-M. and M.W.; visualization, S.A.-O.-M., M.W., O.K. and Z.R.; supervision, S.A.-O.-M., M.W., O.K. and Z.R.; project administration, S.A.-O.-M., O.K. and Z.R. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: As part of the research methodology of the article, a questionnaire survey was conducted, which the respondents filled out anonymously. The Ethics Committee or Institutional Review Board does not apply to the said survey.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: The data used in this article is stored in a university environment. Further information may be sought from the corresponding author.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A. Demographic and Professional Profile of 1000 Survey Respondents

The sample demographics show that 52.6% of the sample is male and 47.4% is female. The data concerning the age of the sample studied indicate that 3.6% of policyholders are between 18 and 24 years old, that 50.3% are between 25 and 34 years old, that 28.7% are between 35 and 44 years old, that 14.4% are between 45 and 55 years old, and that 3% are over 55 years old. These frequencies reveal that more than half of the sample is made up of young people under the age of 35 (Table A1).

Table A1. Sex of policyholder respondents.

	Respondents	Percentage	Cumulative Percentage
Female	474	47.4%	47.4%
Male	526	52.6%	100%
Total	1000	100%	-

Out of the 1000 subjects, 394 or 34.9% of the subjects come from the city of Rabat, 311 or 31.1% reside in the city of Salé, and 340 or 34% live in the city of Kénitra. As for the socio-professional categories, the sample includes 76 executives, or 7.6% of the total; 207 subjects employed in liberal professions, or 20.7% of the total; 613 employees, or 61.3% of the total; 59 students, or 5.9% of the total; and 45 people practicing other professions than those mentioned above (Table A2). More than 60% of those surveyed are employees with more or less fixed incomes, possessing an inflexible purchasing power in relation to positive fluctuations in inflation rates and pricing policies of price increases.

Table A2. Survey respondent socio-professional categories.

	Respondents	Percentage	Cumulative Percentage
Executives	76	7.6%	7.6%
Professions	207	20.7%	28.3%
Employees	613	61.3%	89.6%
Students	59	5.9%	95.5%
Others	45	4.5%	100%
Total	1000	100%	-

Appendix B. Binary Logistic Regression: Logit Transformation

A sample of (n) policyholders belonging to the automobile insurance sector was analyzed to assess the success of CSR among automobile insurance service providers, as explained by a set of digital determinants.

The column vector = () was used, which accounts the dichotomous response variable reflecting the success or failure of the CSR concept among automobile insurers. In addition, we consider (*p*) independent qualitative digital determinants, organized in the matrix (*X*) = (). The column vector β , of dimension (*p*), contains the unknown parameters of the model, namely, the regression coefficients. The modeling of the dichotomous response variable (which relies on the statistical method of binary logistic regression $y_iy_1, y_2, \ldots, y_nX_1, X_2, \ldots, X_py_i$) was performed as follows:

$$\text{Logit}(\pi) = \ln = \left(\frac{\pi}{1-\pi}\right) \sum_{k=0}^{p} \beta_k x_{ik}, \text{ With } i = 1, \dots, n$$
(A1)

Via Logit transformation, we obtain Equation (A2) from Equation (A1):

$$\left(\frac{\pi}{1-\pi}\right) = \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right) \tag{A2}$$

Equation (A2) is evaluated to obtain π and $1 - \pi$:

$$\pi = \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right) - \pi \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right)$$
(A3)

$$\pi + \pi \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right) = \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right)$$
(A4)

$$\pi \left(1 + \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right) \right) = \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right)$$
(A5)

$$\pi = \left(\frac{\exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right)}{1 + \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right)}\right)$$
(A6)

$$\pi = \left(\frac{1}{1 + \exp\left(-\sum_{k=0}^{p} \beta_k x_{ik}\right)}\right) \tag{A7}$$

Similarly, $(1 - \pi)$ is obtained as follows:

$$1 - \pi = 1 - \left(\frac{1}{1 + \exp\left(-\sum_{k=0}^{p} \beta_k x_{ik}\right)}\right)$$
(A8)

$$1 - \pi = \left(\frac{1}{1 + \exp\left(\sum_{k=0}^{p} \beta_k x_{ik}\right)}\right)$$
(A9)

$$1 - \pi = \frac{\exp\left(-\sum_{k=0}^{p} \beta_k x_{ik}\right)}{1 + \exp\left(-\sum_{k=0}^{p} \beta_k x_{ik}\right)}$$
(A10)

Appendix C. Estimation of the β Parameters of the Nonlinear Equations of the Bernoulli Distribution Using the Maximum Likelihood Estimator (MLE)

The binarity of the response variable y_i requires that it must necessarily take two values: 0 or 1. Let y_i represent the probability of the success or failure of the CSR concept among insurers. This probability, noted as π , thus signifies the success of the CSR concept, given the independent variables X, expressed as P ($y_i = 1 \mid X$). Conversely, $1 - \pi$ delimits that y_i which is equal to 0, thus indicating the failure of the CSR concept, expressed by P

 $(y_i = 0 | X)$. When $y_i = 1$, the contribution to the likelihood function is π ; when $y_i = 0$, the contribution is $1 - \pi$. Consequently, the likelihood function is formulated as follows:

$$\pi^{y_i} \ (1-\pi)^{1-y_i} \tag{A11}$$

At this point, the maximum likelihood estimator (MLE) is used to estimate the (p + 1) unknown parameters β as follows:

$$L(y_1, y_2, \dots, y_n, \pi) = \prod_{i=1}^n \pi^{y_i} (1-\pi)^{1-y_i}$$
(A12)

The maximum likelihood method is introduced at this stage to estimate the values of the unknown regression parameters β to maximize the probability of the success of the CSR concept:

$$L(y_1, y_2, \dots y_n, \pi) = \prod_{i=1}^n \pi^{y_i} (1-\pi)^{1-y_i}$$
(A13)

$$=\prod_{i=1}^{n} \left(\frac{\pi}{1-\pi}\right)^{y_i} (1-\pi)$$
(A14)

After changing the first term of Equation (A2) and the second term of Equation (A8), we obtain the following:

$$L(y_{1}, y_{2}, \dots, y_{n}, \beta_{1}, \beta_{2}, \dots, \beta_{p},) = \prod_{i=1}^{n} \left(\exp\left(\sum_{k=0}^{p} \beta_{k} x_{ik}, \right) \right)^{y_{i}} \left(1 - \frac{\exp\left(\sum_{k=0}^{p} \beta_{k} x_{ik}\right)}{1 + \exp\left(\sum_{k=0}^{p} \beta_{k} x_{ik}\right)} \right)$$
(A15)

So,

$$L(y_1, y_2, \dots, y_n, \beta_1, \beta_2, \dots, \beta_p) = \prod_{i=1}^n \left(\exp\left(y_i \sum_{k=0}^p \beta_k x_{ik}\right) \right) \left(1 + \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right) \right)^{-1}$$
(A16)

To simplify the calculations, the logarithmic function of the likelihood function is introduced as shown below:

$$\ln(L(y_1, y_2, \dots, y_n, \beta_1, \beta_2, \dots, \beta_p)) = \ln\left(\prod_{i=1}^n \left(\exp\left(y_i \sum_{k=0}^p \beta_k x_{ik}\right)\right) \left(1 + \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right)\right)^{-1}\right)$$
(A17)

$$\uparrow(y_1, y_2, \dots, y_n, \beta_1, \beta_2, \dots, \beta_p) = \sum_{i=1}^n y_i \left(\sum_{k=0}^p \beta_k x_{ik} \right) - \ln \left(1 + \exp \left(\sum_{k=0}^p \beta_k x_{ik} \right) \right) \tag{A18}$$

Deriving the last equation from the natural logarithm of the likelihood function above, the following is obtained:

$$\frac{\partial \widehat{\downarrow}(\beta)}{\partial \beta_k} = \sum_{i=1}^n y_i \, x_{ik} - \frac{1}{1 + \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right)} \times \frac{\partial}{\partial \beta_k} \left(1 + \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right)\right) \tag{A19}$$

$$\frac{\partial \uparrow(\beta)}{\partial \beta_k} = \sum_{i=1}^n y_i \, x_{ik} - \frac{1}{1 + \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right)} \times \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right) \times \frac{\partial}{\partial \beta_k} \sum_{k=0}^p \beta_k x_{ik} \tag{A20}$$

$$\frac{\partial \uparrow(\beta)}{\partial \beta_k} = \sum_{i=1}^n y_i \, x_{ik} - \frac{x_{ik}}{1 + \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right)} \times \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right) \tag{A21}$$

We know that

$$\frac{\partial}{\partial \beta_k} = \sum_{k=0}^p \beta_k x_{ik} = x_{ik} \tag{A22}$$

So,

$$\frac{\partial \uparrow(\beta)}{\partial \beta_k} = \uparrow'_{\beta_k} = \sum_{i=1}^n y_i \, x_{ik} - \pi . x_{ik} \tag{A23}$$

Estimating the parameters $\hat{\beta} = (\hat{\beta}_0, \hat{\beta}_1, \dots, \hat{\beta}_p)$ that maximize the log-likelihood function (\uparrow) is achievable by setting the (p + 1) equations of (\uparrow') (the gradient of \uparrow') to zero, as shown in Equation (A12), and ensuring that its Hessian matrix (\uparrow'') is negative definite, i.e., every element on the diagonal of this matrix is less than zero.

$$\frac{\partial^2 \uparrow(\beta)}{\partial \beta_k \partial \beta_{k'}} = \frac{\partial}{\partial \beta_{k'}} \sum_{i=1}^n y_i \, x_{ik} - \pi. \, x_{ik} \tag{A24}$$

$$\frac{\partial^2 \uparrow(\beta)}{\partial \beta_k \partial \beta_{k'}} = \frac{\partial}{\partial \beta_{k'}} (-\pi, x_{ik})$$
(A25)

$$\frac{\partial^2 \uparrow(\beta)}{\partial \beta_k \partial \beta_{k'}} = -x_{ik} \frac{\partial}{\partial \beta_{k'}} \left(\frac{\exp\left(\sum_{k=0}^p \beta_k x_{ik}\right)}{1 + \exp\left(\sum_{k=0}^p \beta_k x_{ik}\right)} \right)$$
(A26)

$$\zeta_{\beta_{k\beta'_k}}^{\prime\prime} = -x_{ik}\pi(1-\pi) \ x_{ik}$$
(A27)

Estimating the parameters $\hat{\beta} = (\hat{\beta}_0, \hat{\beta}_1, \dots, \hat{\beta}_p)$ requires the use of the Newton–Raphson iterative optimization method. This method consists of starting the search with an initial value denoted β^0 or β^{old} . The result of this algorithm in matrix notation is written as follows:

$$\beta^{new} = \beta^{old} + \left[- \uparrow'' \left(\beta^{old} \right) \right]^{-1} \times \uparrow' \left(\beta^{old} \right)$$

Appendix D. Hypotheses Testing

Reliability testing was performed as follows: the reliability test marked a value of the "Cronbach's Alpha" coefficient $\hat{\alpha} = 0.881$, far exceeding the minimum conventional threshold of $\hat{\alpha} = 0.70$ (George and Mallery 2003). This confirmed that for this assortment of digital determinants, composed of six elements (Table A3), satisfactory internal consistency was achieved.

Table A3. The digital determinants associated with the success of the concept of CSR in the automobile insurance sector.

Digital Determinants	Code
Claims management platforms and online customer portals	PGL
Digital roadside assistance	ARN
Digital vehicle assessment and inspection	ENV
Advanced security technologies	TSA
Data analysis for personalization	ADP
Augmented reality (AR) and virtual reality (VR) for enhanced customer engagement	RAV
Response Variable	Notation
Success of CSR in the automobile insurance sector	Y = 1
Failure of CSR in the automobile insurance sector	Y = 0

Area-under-the-curve (AUC) analysis, widely used to measure the accuracy of diagnostic tests, expresses the probability of placing a positive element in front of a negative element (Figure A1). However, this technique proposes an AUC = 0.5 as a reference point that must be exceeded. At first glance, all results are highly significant, with a p = 0.000 ≤ 0.05 . Additionally, the table also reports AUCs that exceed the benchmark threshold (AUC = 0.5), i.e., the explanatory variables used in the model all have a significant impact on the response variable. At this stage, claims management platforms and online customer portals (PGL) have a probability of 71.6% vis-à-vis stimulating the improvement of the uptake of the concept of corporate social responsibility (CSR) among motor insurance providers with regard to their policyholders. Likewise, digital road assistance (ARN) has a probability of 66.8% in terms of increasing the adoption of this social notion. Additionally, digital vehicle assessment and inspection (ENV), advanced safety technology (TSA), data analytics for personalization (ADP), and augmented reality (AR) and virtual reality (VR) (RAV), respectively, present probabilities of 64.1%, 79.1%, 61.7%, and 62.6% in terms of enhancing the uptake of the concept of CSR in the automobile insurance industry and consolidating the interrelations between policyholders and their insurance providers.



Figure A1. The AUCs of the different digital determinants associated with CSR success.

In Figure A1, the red and blue lines represent the performance of the numerical determinants in the ROC (Receiver Operating Characteristic) analysis to predict the success of CSR (Corporate Social Responsibility) in the sector of car insurance. The red line corresponds to the reference line with an AUC (Area Under Curve) of 0.5, which represents the performance threshold of a random model, that is to say without discrimination capacity. This line is used as a reference point to evaluate the predictive performance of the models. The blue line shows the actual performance of each numerical determinant tested. If the blue curve is to the right of the red line, it means that the numerical determinant has significant predictive ability, with an AUC above 0.5, indicating that it effectively contributes to predicting CSR success. The further the blue curve extends beyond the red line, the stronger the predictive capacity of the determinant. In summary, Figure A1 illustrates that all the numerical determinants tested exceed the threshold of 0.5 for the AUC, thus demonstrating a significant contribution of each of them to the prediction of CSR success.

According to the chi-square test, the connection between digital determinants, online claims management platforms and customer portals, digital roadside assistance, digital vehicle assessment and inspection, advanced safety technologies, data analysis for personalization, augmented reality (AR) and virtual reality (VR), and the response variable "the success of the concept of CSR" in the automobile insurance industry is highly significant, with an asymptotic significance (two-sided) of p = 0.000 < 0.05. These results led the research team to reject the null hypothesis H_0. The explanatory determinants chosen in this study had a largely significant link with the response variable, i.e., a significant influence

on the success of CSR in the automobile insurance branches and the consolidation of the insured–insurer relationship.

Regression coefficients/equation variables: the results suggest that all the predictor variables and digital determinants have a highly significant effect on the response variable "the success of CSR" in the automobile insurance branch with regard to policyholders.

References

- Abdi, Hervé, and Dominique Valentin. 2007. Multiple Correspondence Analysis. *Encyclopedia of Measurement and Statistics* 2: 651–57. Adwere-Boamah, Joseph, and Shirley Hufstedler. 2015. Predicting Social Trust with Binary Logistic Regression. *Research in Higher*
- Education Journal 27: EJ1056177.
- Ahmad, Maqsood, Qiang Wu, and Muhammad Sualeh Khattak. 2023. Intellectual capital, corporate social responsibility and sustainable competitive performance of small and medium-sized enterprises: Mediating effects of organizational innovation. *Kybernetes* 52: 4014–40. [CrossRef]
- Ahmad, Maqsood, Qiang Wu, and Shakeel Ahmed. 2024. Does CSR digitalization improve the sustainable competitive performance of SMEs? Evidence from an emerging economy. *Sustainability Accounting, Management and Policy Journal* 15: 119–47. [CrossRef]
- Albrecher, Hansjörg, Antoine Bommier, Damir Filipović, Pablo Koch-Medina, Stéphane Loisel, and Hato Schmeiser. 2019. Insurance: Models, Digitalization, and Data Science. *European Actuarial Journal* 9: 349–60. [CrossRef]
- Alieva, Jamila, and Daryl Powell. 2023. The significance of digital waste in the automation of Lean practices. *Quality Management Journal* 30: 121–34. [CrossRef]
- Asif, Muhammad, Shuang Wang, Muhammad Farrukh Shahzad, and Muhammad Ashfaq. 2024. Data Privacy and Cybersecurity Challenges in the Digital Transformation of the Banking Sector. *Computers & Security* 147: 104051.
- Banerjee, Debashmita, Roshan Kumar, Srishti Tripathi, and Benrithung Murry. 2024. Application of Binary Logistic Regression in Biological Studies. Journal of the Practice of Cardiovascular Sciences 10: 48. [CrossRef]
- Bendell, Jem. 2005. In Whose Name? The Accountability of Corporate Social Responsibility. *Development in Practice* 15: 362–74. [CrossRef]
- Bianchi, Cristian, Rosario Merlino, and Roberto Passerone. 2024. Combining Optimization and Simulation for Next-Generation Off-Road Vehicle E/E Architectural Design. *Sensors* 24: 4889. [CrossRef]
- Bilgihan, Anil, Jay Kandampully, and Tingting Zhang. 2016. Towards a unified customer experience in online shopping environments: Antecedents and outcomes. *International Journal of Quality and Service Sciences* 8: 102–19. [CrossRef]
- Bohnert, Alexander, Albrecht Fritzsche, and Shirley Gregor. 2019. Digital Agendas in the Insurance Industry: The Importance of Comprehensive Approaches. *The Geneva Papers on Risk and Insurance Issues and Practice* 44: 1–19. [CrossRef]
- Carroll, Archie B., and Kareem M. Shabana. 2010. The Business Case for Corporate Social Responsibility: A Review of Concepts, Research and Practice. *International Journal of Management Reviews* 12: 85–105. [CrossRef]
- ChAD. 2021. L'intelligence Artificielle en Assurance de Dommages. Chambre de L'assurance de Dommages. 6 Juillet. Available online: https://chad.ca/actualites/2021/07/reflexion-sur-ladoption-de-lintelligence-artificielle-en-assurance-de-dommages/ (accessed on 23 June 2024).
- Chaudhuri, Ranjan, Sheshadri Chatterjee, Demetris Vrontis, Antonino Galati, and Evangelia Siachou. 2023. Examining the issue of employee intentions to learn and adopt digital technology. *Worldwide Hospitality and Tourism Themes* 15: 279–94. [CrossRef]
- Chen, Tzung-Ming, Jia-Qi Wu, and Jian-Ting Lin. 2023. Classification of 3D Casting Models for Product Lifecycle Management and Corporate Sustainability. *Sustainability* 15: 12683. [CrossRef]
- Chen, Xiaoxia, Mélanie Despeisse, and Björn Johansson. 2020. Environmental sustainability of digitalization in manufacturing: A review. *Sustainability* 12: 10298. [CrossRef]
- Chien, Steve, Jason Swope, Qing Yue, Javier Bosch-Lluis, and William Deal. 2021. Using a Digital Twin Weather Research and Forecasting (WRF) Model for Machine Learning of Deep Convective Ice Storms. Paper presented at AGU Fall Meeting 2021, New Orleans, LA, USA, December 13–17.
- Ciarli, Tommaso, Martin Kenney, Silvia Massini, and Lucia Piscitello. 2021. Digital technologies, innovation, and skills: Emerging trajectories and challenges. *Research Policy* 50: 104289. [CrossRef]
- Cirillo, Valeria, Lucrezia Fanti, Andrea Mina, and Andrea Ricci. 2023. The adoption of digital technologies: Investment, skills, work organisation. *Structural Change and Economic Dynamics* 66: 89–105. [CrossRef]
- DataCamp. 2023. L'impact Environnemental des Technologies Numériques et des Données. DataCamp. Available online: https://www.datacamp.com/blog/environmental-impact-data-digital-technology (accessed on 23 June 2024).
- Donald, Betsy, and Shauna Brail. 2024. *Urban Mobility: How the iPhone, COVID, and Climate Changed Everything*. Toronto: University of Toronto Press. Available online: https://muse.jhu.edu/book/123695 (accessed on 5 April 2024).
- Dudek, Marek, Iryna Bashynska, Svitlana Filyppova, Svitlana Yermak, and Dariusz Cichoń. 2023. Methodology for assessment of inclusive social responsibility of the energy industry enterprises. *Journal of Cleaner Production* 394: 136317. [CrossRef]
- Eckert, Christian, Christof Neunsinger, and Katrin Osterrieder. 2022. Managing Customer Satisfaction: Digital Applications for Insurance Companies. *The Geneva Papers on Risk and Insurance Issues and Practice* 47: 569–602. [CrossRef]
- Eckert, Christian, Johanna Eckert, and Armin Zitzmann. 2021. The Status Quo of Digital Transformation in Insurance Sales: An Empirical Analysis of the German Insurance Industry. Zeitschrift Für Die Gesamte Versicherungswissenschaft 110: 133–55. [CrossRef]

- Ehrentraud, Johannes, Denise Garcia Ocampo, Lorena Garzoni, and Mateo Piccolo. 2020. Policy Responses to Fintech: A Cross-Country Overview. *Policy Commons*. Available online: https://policycommons.net/artifacts/3708795/policy-responses-to-fintech/451476 3/ (accessed on 18 May 2024).
- Eling, Martin, and Martin Lehmann. 2018. The Impact of Digitalization on the Insurance Value Chain and the Insurability of Risks. *The Geneva Papers on Risk and Insurance Issues and Practice* 43: 359–96. [CrossRef]
- Eling, Martin, Davide Nuessle, and Julian Staubli. 2022. The Impact of Artificial Intelligence along the Insurance Value Chain and on the Insurability of Risks. *The Geneva Papers on Risk and Insurance Issues and Practice* 47: 205–41. [CrossRef]
- Erem Ceylan, Işıl. 2022. The Effects of Artificial Intelligence on the Insurance Sector: Emergence, Applications, Challenges, and Opportunities. In *The Impact of Artificial Intelligence on Governance, Economics and Finance, Volume 2*. Edited by Sezer Bozkuş Kahyaoğlu. Singapore: Springer Nature, pp. 225–41. [CrossRef]
- Esposito, Paolo, and Paolo Ricci. 2021. Cultural Organizations, Digital Corporate Social Responsibility and Stakeholder Engagement in Virtual Museums: A Multiple Case Study. How Digitization Is Influencing the Attitude toward CSR. *Corporate Social Responsibility and Environmental Management* 28: 953–64. [CrossRef]
- Feroz, Abdul Karim, Hangjung Zo, and Ananth Chiravuri. 2021. Digital transformation and environmental sustainability: A review and research agenda. *Sustainability* 13: 1530. [CrossRef]
- Garde, M. D., and Gautam Prasad. 2018. Case Study: Motor Claims Management. *Bimaquest*. 18. Available online: http://www. bimaquest.niapune.org.in/index.php/bimaquest/article/view/44 (accessed on 23 February 2024).
- Ge, Wenxia, and George Alex Whitmore. 2010. Binary response and logistic regression in recent accounting research publications: A methodological note. *Review of Quantitative Finance and Accounting* 34: 81–93. [CrossRef]
- George, Darren, and Paul Mallery. 2003. Cronbach's Alpha. SPSS for Windows Step by Step: A Simple Guide and Reference. 11.0 Update. January. Available online: https://www.researchgate.net/publication/234827666_SPSS_for_Windows_Step-by-Step_A_Simple_ Guide_and_Reference_140_update_7th_Edition (accessed on 19 June 2024).
- Gillis, Steven. 2023. Blockchain-Based Application for Insurance Claims Management. Available online: https://dash.lib.harvard.edu/ handle/1/37375031 (accessed on 21 May 2024).
- Goyal, Nupur, Sanjay Singh Chauhan, Shweta Pandey, Rajesh Singh, Anil Kumar, and Kapil Joshi. 2023. Financial Services 4.0—Future Perspective based on Mixed Reality. Paper presented at 2023 IEEE International Conference on Contemporary Computing and Communications (InC4), Bangalore, India, April 21–22, vol. 1, pp. 1–4. [CrossRef]
- Haji, Abdifatah Ahmed, and Nazli Anum Mohd Ghazali. 2018. The role of intangible assets and liabilities in firm performance: Empirical evidence. *Journal of Applied Accounting Research* 19: 42–59. [CrossRef]
- Hanelt, André, René Bohnsack, David Marz, and Cláudia Antunes Marante. 2021. A Systematic Review of the Literature on Digital Transformation: Insights and Implications for Strategy and Organizational Change. *Journal of Management Studies* 58: 1159–97. [CrossRef]
- Heeks, Richard. 2002. Information Systems and Developing Countries: Failure, Success, and Local Improvisations. *The Information Society* 18: 101–12. [CrossRef]
- Hernández, Ana Cecilia Chumaceiro, Judith García de Velazco, Ivana V. Reyes Hernández, and Rafael Ravina Ripoll. 2020. University Social Responsibility in the Organizational Happiness Management. *Utopía y Praxis Latinoamericana* 25: 427–440.
- Hirsch-Kreinsen, Hartmut. 2020. Digitale Transformation von Arbeit: Entwicklungstrends und Gestaltungsansätze. Stuttgart: Kohlhammer Verlag.
- Hofmann, Bjørn. 2022. Open Science Knowledge Production: Addressing Epistemological Challenges and Ethical Implications. *Publications* 10: 24. [CrossRef]
- IndustryWired. 2021. How AI Is Changing Personal Injury Claims Images. *IndustryWired*. (blog). December 20. Available online: https://industrywired.com/how-ai-is-changing-personal-injury-claims-images/ (accessed on 5 April 2024).
- Ismail Abdelaal, Mariam Helmy, Mohamed Khater, and Mohamed Zaki. 2018. Digital Business Transformation and Strategy: What Do We Know So Far? Working Paper. *Cambridge Service Alliance* 10: 1–35. [CrossRef]
- Janiszewska-Kiewra, Ewa, Jannik Podlesny, and Henning Soller. 2024. Utilisation éthique des Données à L'ère du Numérique et de la Réglementation. Consulté le September 11. Available online: https://www.mckinsey.com/capabilities/mckinsey-digital/our-insights/tech-forward/ethical-data-usage-in-an-era-of-digital-technology-and-regulation (accessed on 29 July 2024).
- Kavitha, N. V., and T. Anuradha. 2016. Corporate Social Responsibility in Insurance Sector in India. International Journal of Innovative Research and Development 5: 137–43. Available online: https://www.internationaljournalcorner.com/index.php/ijird_ojs/article/ view/136074 (accessed on 18 March 2024).
- Kelley, Kevin H., Lisa M. Fontanetta, Mark Heintzman, and Nikki Pereira. 2018. Intelligence Artificielle: Implications pour L'inflation Sociale et L'assurance-Kelley-2018-Risk Management and Insurance Review-Bibliothèque en Ligne Wiley. Available online: https://onlinelibrary.wiley.com/doi/abs/10.1111/rmir.12111 (accessed on 15 September 2024).
- Kemboi, Joan J. 2022. Effect of Claims Digitalization on Service Delivery by Insurance Companies in Kenya. Master of Business Administration thesis, University of Nairobi, Nairobi, Kenya. Available online: http://erepository.uonbi.ac.ke/handle/11295/16 3496 (accessed on 12 October 2024).
- Khovrak, Inna. 2020. ESG-Driven Approach to Managing Insurance Companies' Sustainable Development. Insurance Markets and Companies 11: 42–52. [CrossRef]

- Kumar, Sharath, and Vikram Gota. 2023. Logistic regression in cancer research: A narrative review of the concept, analysis, and interpretation. *Cancer Research, Statistics, and Treatment* 6: 573. [CrossRef]
- Leukel, Joerg, Gülistan Özbek, and Vijayan Sugumaran. 2024. Application of Logistic Regression to Explain Internet Use among Older Adults: A Review of the Empirical Literature. *Universal Access in the Information Society* 23: 621–35. [CrossRef]
- Liu, Huifang, and Jin-Sup Jung. 2021. The Effect of CSR Attributes on CSR Authenticity: Focusing on Mediating Effects of Digital Transformation. *Sustainability* 13: 7206. [CrossRef]
- Lustman, Florence. 2021. Crise de la COVID-19: La place de l'assurance dans le monde d'après. *Revue D'économie Financière* 139–140: 61–67. [CrossRef]
- Martínez-Caro, Eva, Juan Gabriel Cegarra-Navarro, and Francisco Javier Alfonso-Ruiz. 2020. Digital technologies and firm performance: The role of digital organisational culture. *Technological Forecasting and Social Change* 154: 119962. [CrossRef]
- Mattioli, Veronica, Luca Davoli, Laura Belli, Sara Gambetta, Luca Carnevali, Andrea Sgoifo, Riccardo Raheli, and Gianluigi Ferrari. 2024. IoT-Based Assessment of a Driver's Stress Level. *Sensors* 24: 5479. [CrossRef] [PubMed]
- Merrill, John P., Eric J. Suchomel, Srinivasa Varadhan, Melissa Asher, Lea Z. Kane, Elisabeth L. Hawley, and Rula A. Deeb. 2019. Development and Validation of Technologies for Remediation of 1,2,3-Trichloropropane in Groundwater. *Current Pollution Reports* 5: 228–37. [CrossRef]
- Mizgier, Kamil J., Otto Kocsis, and Stephan M. Wagner. 2018. Zurich Insurance Uses Data Analytics to Leverage the BI Insurance Proposition. *Interfaces* 48: 94–107. [CrossRef]
- Muiru, Harrison Mugo. 2024. Effects of Payer-Provider Automation Technologies on Operational Performance of Medical Insurance Providers in Nairobi City County, Kenya. Strathmore University. Available online: https://su-plus.strathmore.edu/items/e400f6 6a-3330-410c-904a-5612a85b78a3 (accessed on 16 July 2024).
- Murray, Alex, Scott Kuban, Matt Josefy, and Jon Anderson. 2021. Contracting in the Smart Era: The Implications of Blockchain and Decentralized Autonomous Organizations for Contracting and Corporate Governance. Academy of Management Perspectives 35: 622–41. [CrossRef]

Nicoletti, Bernardo. 2020. Insurance 4.0: Benefits and Challenges of Digital Transformation. Berlin and Heidelberg: Springer Nature.

- Nicoletti, Bernardo. 2021. Platforms for Insurance 4.0. In *Insurance 4.0, Palgrave Studies in Financial Services Technology*. Cham: Palgrave Macmillan. [CrossRef]
- Oyewole, Adedoyin Tolulope, Omotayo Bukola Adeoye, Wilhelmina Afua Addy, Chinwe Chinazo Okoye, Onyeka Chrisanctus Ofodile, and Chinonye Esther Ugochukwu. 2024. Augmented and virtual reality in financial services: A review of emerging applications. *World Journal of Advanced Research and Reviews* 21: 551–67. [CrossRef]
- Pangalos, George. 2023. Financing for a Sustainable Dry Bulk Shipping Industry: What Are the Potential Routes for Financial Innovation in Sustainability and Alternative Energy in the Dry Bulk Shipping Industry? *Journal of Risk and Financial Management* 16: 101. [CrossRef]
- Park, Yoon Soo, Lars Konge, and Anthony R. Artino, Jr. 2020. The Positivism Paradigm of Research. *Academic Medicine* 95: 690. [CrossRef] [PubMed]
- Popovic, D., C. Avis, M. Byrne, C. Cheung, M. Donovan, Y. Flynn, C. Fothergill, Z. Hosseinzadeh, Z. Lim, and J. Shah. 2020. Understanding blockchain for insurance use cases. *British Actuarial Journal* 25: e12. [CrossRef]
- Porter, Michael E., and E. James Heppelmann. 2017. Why Every Organization Needs an Augmented Reality Strategy. *Harvard Business Review* 95: 46–57.
- Pratt, Mary K. 2023. What Is a Digital Transformation Strategy? Everything You Need to Know. TechTarget/SearchCIO. Available online: https://www.techtarget.com/searchcio/tip/What-is-a-digital-transformation-strategy-Everything-you-need-to-know (accessed on 18 June 2024).
- Quach, Sara, Park Thaichon, Kelly D. Martin, Scott Weaven, and Robert W. Palmatier. 2022. Digital Technologies: Tensions in Privacy and Data. *Journal of the Academy of Marketing Science* 50: 1299–323. [CrossRef]
- Reis, João, Marlene Amorim, Nuno Melão, and Patrícia Matos. 2018. Digital Transformation: A Literature Review and Guidelines for Future Research. In *Trends and Advances in Information Systems and Technologies*. Edited by Álvaro Rocha, Hojjat Adeli, Luís Paulo Reis and et Sandra Costanzo. Cham: Springer International Publishing, pp. 411–21. [CrossRef]
- Ren, Zhiyuan, and Chen Chen. 2024. Special Issue: Recent Advances in Intelligent Vehicular Networks and Communications. *Electronics* 13: 3096. [CrossRef]
- Richert, Maria, and Marek Dudek. 2023. Risk Mapping: Ranking and Analysis of Selected, Key Risk in Supply Chains. *Journal of Risk and Financial Management* 16: 71. [CrossRef]
- Ross, Kevin. 2020. Quoi de Neuf Avec L'IA (Intelligence Artificielle) dans le Traitement des Demandes D'indemnisation des Accidents du Travail? *Insurance Advocate*. August 15. Available online: https://www.insurance-advocate.com/2020/08/15/whats-newwith-ai-artificial-intelligence-in-workers-compensation-claims-processing/ (accessed on 23 April 2024).
- Ruiter, B. M. 2022. CSR Within Insurance Companies in Europe. Bachelor's thesis, University of Twente, Enschede, The Netherlands, June 28. Available online: https://essay.utwente.nl/90977/ (accessed on 14 June 2024).
- Salvi, Antonio, Stefano Vitolla, Pasquale Raimo, and Mario Rubino. 2021. Online Information on Digitalisation Processes and Its Impact on Firm Value. *Journal of Business Research* 124: 437–444. [CrossRef]

- SAP. 2024. Digitisation vs. Digitalization. Available online: https://www.sap.com/uk/products/erp/digitization-vs-digitalization. html (accessed on 15 September 2024).
- Schaltegger, Stefan, and Marcus Wagner. 2011. Sustainable Entrepreneurship and Sustainability Innovation: Categories and Interactions. Business Strategy and the Environment 20: 222–37. [CrossRef]
- Schaltegger, Stefan, Florian Lüdeke-Freund, and Erik G. Hansen. 2016. Business Models for Sustainability: A Co-Evolutionary Analysis of Sustainable Entrepreneurship, Innovation, and Transformation. *Organization & Environment* 29: 264–89. [CrossRef]
- Soni, Sheetal, Usha Yadav, and Abhishek Soni. 2022. Virtual Reality & Augmented Reality: A way to Digital Transformation of Customer Engagement. Paper presented at 2022 International Conference on Machine Learning, Big Data, Cloud and Parallel Computing (COM-IT-CON), Faridabad, India, May 26–27, vol. 1, pp. 573–77. Available online: https://ieeexplore.ieee.org/ abstract/document/9850954/ (accessed on 12 April 2024).
- Stanly, Anupa, and K. Aruna. 2024. Unlocking Potential: Improving Insurtech Innovation and Efficiency with a Data Analytics Approach. August 27. Available online: https://ssrn.com/abstract=4938167 (accessed on 24 August 2024).
- Stewart, Harrison. 2023. Digital Transformation Security Challenges. Journal of Computer Information Systems 63: 919–36. [CrossRef]
- Stoeckli, Emanuel, Christian Dremel, and Falk Uebernickel. 2018. Exploring Characteristics and Transformational Capabilities of InsurTech Innovations to Understand Insurance Value Creation in a Digital World. *Electronic Markets* 28: 287–305. [CrossRef]
- Teece, David J. 2018. Dynamic capabilities as (workable) management systems theory. *Journal of Management & Organization* 24: 359–68. [CrossRef]
- Teece, David J., Gary Pisano, and Amy Shuen. 1997. Dynamic capabilities and strategic management. *Strategic Management Journal* 18: 509–33. [CrossRef]
- The United Nations Environment Programme. 2021. How Digital Technology and Innovation Can Help Protect the Planet. November 23. Available online: https://www.unep.org/news-and-stories/story/how-digital-technology-and-innovation-can-help-protect-planet (accessed on 12 January 2024).
- Trescases, Anne. 2019. Les plateformes: Assureurs du XXIe siècle? Revue Internationale de droit Économique 3: 291–304. [CrossRef]
- Van Alstyne, Marshall W., and Geoffrey G. Parker. 2021. Digital Transformation Changes How Companies Create Value. Harvard Business Review. Available online: https://hbr.org/2021/12/digital-transformation-changes-how-companies-create-value (accessed on 12 July 2024).
- van Tonder, Estelle, and Daniël Johannes Petzer. 2018. The interrelationships between relationship marketing constructs and customer engagement dimensions. *The Service Industries Journal* 38: 948–73. [CrossRef]
- Van Veldhoven, Ziboud, and Jan Vanthienen. 2023. Best Practices for Digital Transformation Based on a Systematic Literature Review. Digital Transformation and Society 2: 104–28. [CrossRef]
- Varshney, Deepanjana. 2020. Digital Transformation and Creation of an Agile Workforce: Exploring Company Initiatives and Employee Attitudes. In Contemporary Global Issues in Human Resource Management. Edited by Mehmet Ali Turkmenoglu et Berat Cicek. Bingley: Emerald Publishing Limited, pp. 89–105. [CrossRef]
- Verhoef, Peter C., Thijs Broekhuizen, Yakov Bart, Abhi Bhattacharya, John Qi Dong, Nicolai Fabian, and Michael Haenlein. 2021. Digital transformation: A multidisciplinary reflection and research agenda. *Journal of Business Research* 122: 889–901. [CrossRef]
- Verma, Deepak, and Madhu Bala. 2018. A Critical Review of Digital Marketing. International Journal of Management, IT & Engineering 8: 321–39.
- Voigt, Paul, and Axel Von Dem Bussche. 2017. *The EU General Data Protection Regulation (GDPR)*. Cham: Springer International Publishing. [CrossRef]
- Wagner, Beverly. 2015. Implementing and Managing Economic, Social and Environmental Efforts of Business Sustainability. Bingley: Emerald Group Publishing Limited. [CrossRef]
- Wang, Fenghua, Yuhan Jia, Guanwei Li, Monica Lam, and Ye Liu. 2024. An Empirical Study of the Relationship between Digital Transformation, Corporate Social Responsibility, and Financial Performance. *Business Ethics and Leadership* 8: 57–73. [CrossRef]
- Wiegand, Tina, and Martin Wynn. 2023. Sustainability, the circular economy and digitalisation in the German textile and clothing industry. *Sustainability* 15: 9111. [CrossRef]
- Wieland, Désirée A. C., Björn S. Ivens, Elizaveta Kutschma, and Philipp A. Rauschnabel. 2024. Augmented and Virtual Reality in Managing B2B Customer Experiences. *Industrial Marketing Management* 119: 193–205. [CrossRef]
- Wynn, Martin G. 2021. Handbook of Research on Digital Transformation, Industry Use Cases, and the Impact of Disruptive Technologies. Hershey: IGI-Global. Available online: https://www.igi-global.com/book/digital-transformation-industry-use-cases/265448 (accessed on 12 April 2024).
- Wynn, Martin G., and Kerstin Felser. 2023. Digitalisation and Change in the Management of IT. Computers 12: 251. [CrossRef]
- Wynn, Martin G., and Peter Jones. 2023. Corporate Responsibility in the Digital Era. Information 14: 324. [CrossRef]
- Xin, Daleng, Yanzhen Yi, and Jianjun Du. 2022. Does Digital Finance Promote Corporate Social Responsibility of Pollution-Intensive Industry? Evidence from Chinese Listed Companies. *Environmental Science and Pollution Research* 29: 85143–59. [CrossRef]
- Yokoi, Tomoko, Lazaros Goutas, Michael Wade, and Nicolas Zahn. 2023. Infusing Digital Responsibility into Your Organization. *Harvard Business Review*. April 28. Available online: https://hbr.org/2023/04/infusing-digital-responsibility-into-your-organization?ab= hero-subleft-2 (accessed on 12 July 2024).

Yun, Hyukchoon, and Stephane Barde. 2024. A Novel Concept for Integrating and Delivering Automobile Warranty Reliability Information via a Comprehensive Digital Dashboard. Available online: https://ssrn.com/abstract=4781836 (accessed on 10 September 2024).

Zhang, Meixuan, and Zongsheng Huang. 2024. The Impact of Digital Transformation on ESG Performance: The Role of Supply Chain Resilience. *Sustainability* 16: 7621. [CrossRef]

Zhang, Minglong, and Yin Liu. 2022. Influence of digital finance and green technology innovation on China's carbon emission efficiency: Empirical analysis based on spatial metrology. *Science of The Total Environment* 838: 156463. [CrossRef] [PubMed]

Zuperkienė, Erika, Ligita Šimanskienė, Mariantonietta Fiore, and Julius Paulikas. 2023. Managing Employee Resistance to Technological Change for Organizational Sustainability. In *Technology, Business, and Sustainability*. New York: Routledge.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.